Author's response to reviews

Title: A methodological approach to identify external factors for indicator-based risk adjustment illustrated by a cataract surgery register

Authors:

Ursula Hahn (U.Hahn@t-online.de)
Irmingard Neuhann (Dr.I.Neuhann@neuhann.de)
Stephanie Schmickler (st.schmickler@augen-zentrum-nordwest.de)
Frank Krummenauer (Frank.Krummenauer@uni-wh.de)

Version: 3
Date: 19 November 2013

Author's response to reviews:

Answer to your Mail from 23.Octoberly 2013

Dear Sirs,

thank you for your formal revision of our submission MS: 3708139201027085.

Your comment: We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.

Authors: A native English speaker has revised the manuscript and the abstract.

Your comment: Please provide a link or reference to the relevant local legislation that states that ethical approval is not required for this study in the 'Methods' section of the manuscript and also indicate which body or authority gave permission to use the data presented in the manuscript.

Authors: We comment on the German data protection law related to the data presented in the manuscript and provide the requested reference in the method section.

Best regards

Dr. U. Hahn